Author's response to reviews

Title: Epithelioid hemangioma (angiolymphoimmun hyperplasia with eosinophilia) of the orbit: a case report

Authors:

Bruno F Fernandes (bruno.mtl@gmail.com)
Abdullah Al-Mujaini (almujaini@hotmail.com)
Tina Haliotis (tina.haliotis@mcgill.ca)
Abdullah A Al-Kandari (calemero@hotmail.com)
Bryan Arthurs (bryan.arthurs@mcgill.ca)
Miguel N Burnier Jr. (miguel.burnier@mcgill.ca)

Version: 5 Date: 19 June 2007

Author's response to reviews:

Montreal, June 19th, 2007

Dear Dr. Michael Kidd,

Editor-in-Chief
Journal of Medical Case Reports,

Thank you very much for your reviewer’s considerations.

We are glad to have a response and correction regarding our manuscript 1931665201411374, entitled "Orbital involvement in Angiolymphoid Hyperplasia with Eosinophilia: Case Report".

All of the suggestions of the reviewers were attended. The modifications made in our paper are indicated below in bold, after each of the reviewers' suggestions.

Formatting changes requested
-----------------------------
1. Please remove the text ‘title’ the running title, keywords, and all underlining from the text.
   Re: Changes were made.
2. Remove author qualifications e.g. MD PhD.
   Re: Qualifications removed.
3. Put the list of authors on one line and place the emails in a list beneath the corresponding author.
   Re: Changes were made.
3. Legends: -> Figure legends
   Re: Change was made.
4. References - References should be cited in the text using consecutive numbers in square brackets starting from [1] then [2] etc. Please ensure the type of bracket used is consistent. Please do not use superscript numbers.

Re: References were reformatted

Thank you very much for your comments, we concur that those changes enriched our manuscript.

Bruno F. Fernandes,
Abdullah Al-Mujaini;
Tina Petrogiannis-Haliotis,
Abdullah Al-Kandari,
Bryan Arthurs,
Miguel N. Burnier Jr.